Author's response to reviews

Title: Gene expression profiling revealed novel mechanism of action of Taxotere and Furtulon in prostate cancer cells

Authors:

Yiwei Li (yiweili@med.wayne.edu)
Maha Hussain (mhussain@med.umich.edu)
Sarah H Sarkar (ssarkar@med.wayne.edu)
James Eliason (eliasonj@karmanos.org)
Ran Li (ranli@med.wayne.edu)
Fazlul H Sarkar (fsarkar@med.wayne.edu)

Version: 4 Date: 14 January 2005

Author's response to reviews:

January 14, 2005

Iratxe Puebla
Assistant Editor
BioMed Central
Middlesex House
34-42 Cleveland Street
London W1T 4LB

Dear Iratxe Puebla:

Enclosed please find our revised manuscript entitled "Gene expression profiling revealed novel mechanism of action of Taxotere and Furtulon in prostate cancer cells" submitted for publication in the journal of BMC Cancer. Revision has been made according to the comments made by the editor. We will be looking forward for its speedy publication.

Thank you very much.

Sincerely yours,

Fazlul H. Sarkar, Ph.D.
Professor of Pathology
Department of Pathology
Karmanos Cancer Institute
Wayne State University School of Medicine
715 Hudson Webber Cancer Research Center
110 E Warren
Detroit, MI 48201
USA
Tel: 313-966-7279
Fax: 313-966-7558
E-mail: fsarkar@med.wayne.edu

Point-by-point description of the changes made

1. E-mail address: We have added the author's E-mail addresses, which are available.
2. Country: We have added the country in the affiliation details.
3. Page 8: We have given the last paragraph on page 8 the section heading "Conclusions".
4. Competing interests: We have added the section of "Competing interests".
5. Author's contributions: We have added the section of "Author's contributions".
6. Tables: We have added visible black lines on the borders of tables and oriented the tables in portrait.
7. Figure 4: We have deleted the title of figure 4 in the image file.